Author's response to reviews

Title: Hip fracture fixation in the below-knee amputee - a surgical dilemma: A case report.

Authors:
Ulfin Rethnam (ulfinr@yahoo.com)
Rajam SU Yesupalan (ajeesh2000@yahoo.co.uk)
Amer Shoaib (amershoaib@hotmail.com)
Thanga K Ratnam (Tk.ratnam@new-tr.wales.nhs.uk)

Version: 3 Date: 24 August 2007

To
The Editor.
Thank you for the suggestions for revising the manuscript. The following changes have been made to the manuscript:
1. A consent section has been added to the manuscript as per the instructions.
2. The figure legends have been amended.
Please consider the manuscript for publication.
Ulfin Rethnam